Author's response to reviews

Title: Psychostimulant prescribing trends in a paediatric population in Ireland: A national cohort study

Authors:

Fiona Boland (fionaboland@rcsi.ie)
Rose Galvin (rose.galvin@ul.ie)
Udo Reulbach (reulbacu@tcd.ie)
Nicola Motterlini (rose.galvin@ul.ie)
Dervla Kelly (dkelly23@tcd.ie)
Kathleen Bennett (BENNETTK@tcd.ie)
Tom Fahey (tomfahey@rcsi.ie)

Version: 6 Date: 13 August 2015

Author's response to reviews:

Dear Colleagues,

Please find our responses to the editorial revisions suggested by your team:

- Please provide in the title page the email addresses of all authors.
  The email addresses of all authors have been included in the title page of the revised manuscript. The changes have been tracked – see lines 31-38

- Please detail funding in the Acknowledgements section and remove section Funding from the manuscript.
  The separate section on funding has been merged with the acknowledgements section in the revised manuscript. See lines 384-385

- Please provide a List of abbreviations after the Conclusions.
  This has been inserted in the revised manuscript and changes have been tracked – see lines 362-370

- Please place figure titles and legends at the end of the main manuscript and remove them from the figure files.
  The figure titles are now contained in the main manuscript (see lines 518-519 and lines 521-522). The figure legends are embedded in the image generated in the graph. We can remove them if necessary.

Many thanks,

Rose